Author's response to reviews

Title: Neuromuscular training with injury prevention counseling to decrease the risk of acute musculoskeletal injury in young men during military service: A population-based randomized study

Authors:

Jari Parkkari (jari.parkkari@uta.fi)
Henri Taanila (henri.taanila@uta.fi)
Jaana Suni (jaana.h.suni@uta.fi)
Ville M Mattila (ville.mattila@uta.fi)
Olli Ohrankämmen (olli.ohrankammen@mil.fi)
Petteri Vuorinen (petteri.vuorinen@mil.fi)
Pekka Kannus (pekka.kannus@uta.fi)
Harri Pihlajamäki (harri.pihlajamaki@helsinki.fi)

Version: 2 Date: 21 March 2011

Author's response to reviews:

Dear Editor,

We have submitted online our revised manuscript entitled “Neuromuscular training with injury prevention counseling to decrease the risk of acute musculoskeletal injury in young men during military service: A population-based randomized study”. (MS ID : 1725034946513897)

We have now provided the legends for the figures and consent was given by the individuals in the pictures to publish them.

Yours sincerely,

Jari Parkkari, M.D.
UJK Institute, P.O. Box 30
FIN-33501 Tampere, FINLAND
phone: +358-3-282-9111
e-mail: jari.parkkari@uta.fi